Author's response to reviews

Title: Development of a workplace intervention for sick-listed employees with stress-related mental disorders: Intervention Mapping as a useful tool

Authors:

Sandra H van Oostrom (sandra.vanoostrom@vumc.nl)
Johannes R Anema (h.anema@vumc.nl)
Berend Terluin (b.terluin@vumc.nl)
Anita Venema (a.venema@arbeid.tno.nl)
Henrica CW de Vet (hcw.devet@vumc.nl)
Willem van Mechelen (w.vanmechelen@vumc.nl)

Version: 4 Date: 7 August 2007

Author's response to reviews:

Dear Editor,

Thank you for accepting our manuscript entitled 'Development of a workplace intervention for sick-listed employees with stress-related mental disorders: Intervention Mapping as a useful tool' for publication. We have addressed the formatting changes requested by the Editorial Production Team and providing a point-by-point response to inform BMC where and how it the formatting changes are incorporated.

We hope that our manuscript is now suitable for publication.

On behalf of all authors,

Yours sincerely,

S.H. van Oostrom, MSc
EMGO-Institute
VU University Medical Center
Department of Public and Occupational Health
Van der Boechorststraat 7
1081 BT Amsterdam